Author's response to reviews

Title: Older HIV-infected individuals present late and have a higher mortality: a UK cohort study

Authors:

Collins C Iwuji (ciwuji@yahoo.com)
Duncan Churchill (Duncan.Churchill@bsuh.nhs.uk)
Yvonne Gilleece (Yvonne.Gilleece@bsuh.nhs.uk)
Helen Weiss (Helen.Weiss@lshtm.ac.uk)
Martin Fisher (Martin.Fisher@bsuh.nhs.uk)

Version: 2 Date: 13 October 2012

Author's response to reviews:

I have made attended to your comments in the resubmitted paper.

1. I have responded to the question on ethics and consent. Please see text highlighted in yellow in the attached manuscript. I have attached a publication from the UK National Research Ethics Committee referenced in the manuscript for your attention. Relevant sections highlighted in yellow.

Manuscript has also be uploaded through the submission portal

2. Conclusion section added after discussion

3. I have included the the titles for the 3 figures at the end of the manuscript as well as the legend for figure 3.